Author's response to reviews

Title: Toward the Future of Psychiatric Diagnosis: The Seven Pillars of RDoC

Authors:

Bruce N. Cuthbert (bcuthber@mail.nih.gov)
Thomas R. Insel (ti4g@nih.gov)

Version: 4 Date: 11 March 2013

Author's response to reviews:

Thank you for pointing out the appropriate final edits for this manuscript. As requested, we have made the following changes:

1. We added a brief clause in the Abstract regarding the challenges. Please advise if the Abstract is now too long; it will be difficult to reduce the overall Abstract length without omitting some important aspect.

2. We have corrected the references regarding the Philosophy, Ethics, and Humanities in Medicine series.

Sincerely,
Bruce Cuthbert